Author's response to reviews

Title: Translation and validation of the Korean confusion assessment method for the intensive care unit

Authors:

Eun Young Heo (eunyoungheo@gmail.com)
Byoung-Jo Lee (minddoctor@naver.com)
Bong-Jin Hahm (hahm@snu.ac.kr)
Eun Hee Song (julie0315@hanmail.net)
Han-A Lee (hugo0364@hanmail.net)
Chul-Gyu Yoo (cgyoo@snu.ac.kr)
Young Whan Kim (ywkim@snu.ac.kr)
Sung Koo Han (hansk@snu.ac.kr)
Young-Soo Shim (ysshim@snu.ac.kr)
Sang-Min Lee (sangmin2@snu.ac.kr)

Version: 4 Date: 9 May 2011

Author's response to reviews:

Dear Pf. Sabina Alam,

We revised the manuscript entitled as “Translation and validation of the Korean confusion assessment method for the intensive care unit (ID : 1020346906460242)”. based on reviewers’ comments. Corrections were written in red color.

Reviewer 1.
1. Minor Revisions
“typo”? page 7: Every enrolled patient was assessed one than once. Perhaps “more than once”?

We are sorry for the mistake. We corrected it to “more than once” (page 7. Result 2nd paragraph)

Reviewer 2
Minor Essential Revisions
Results; interrater reliability and validity of the Korean CAM-ICU
…enrolled patient was assessed one than once…..
This sentence is incomplete
We corrected the sentence as above.

Throughout the manuscript: please check the tenses (grammar) as they differ
throughout the manuscript.

We are very thankful to your kind comments. We corrected the tenses of several verbs as you recommended (page 6, 7, and 8)

We really appreciate your kindness to give us one more chance for revision of our manuscript.

Sincerely yours,

On behalf of all authors,

Sang-Min Lee, M.D.

Division of Pulmonary and Critical Care Medicine
Department of Internal Medicine
Seoul National University Hospital
101 Daehangno, Jongno-gu, Seoul, 110-744, Republic of Korea

Phone: +82-2-2072-0833
Fax: +82-2-762-9662